Author's response to reviews

Title: Microvessel Changes after Post-ischemic Benign and Malignant Hyperemia: Experimental Study in Rats

Authors:

Haitao Lu (haitaolu1975@gmail.com)
Jungong Zhao (lhtliminghua@gmail.com)
Minghua Li (liminghuapro@gmail.com)
Yingsheng Cheng (chengyingsheng@yahoo.cn)
Yongdong Li (yongdongli@yahoo.cn)
Xiaofang You (youxiaofang@yahoo.cn)
Yuwu Zhao (doctorneurol@yahoo.cn)

Version: 4 Date: 26 March 2010

Author's response to reviews: see over
Ms: 1572415702259995

Title: Microvessel Changes after Post-ischemic Benign and Malignant Hyperemia: Experimental Study in Rats

Dear editor:

Thank you very much for considering the revision of our manuscript. We have re-changed the format of manuscript according to your advice and made a point-by-point response to the reviewer’s comments below.

We appreciate your hard work and hope that our paper has a chance to be published in BMC NEUROLOGY. If you confront any further confusion about the manuscript, please contact me by e-mail: lhtliminghua@gmail.com.

Sincerely

Jungong Zhao MD. and Yuwu Zhao MD.

The Sixth Affiliated People’s Hospital; Medical school of Shanghai Jiao Tong University

E-Mail: lhtliminghua@gmail.com or doctorneurol@yahoo.cn

Tel.: +86-21-64844183

Fax: +86-21-64844183